Author's response to reviews

Title: The important factors in clinical competency of Iranian nurses: A Qualitative Study

Authors:

ali dehghani (ali.dehghani2000@yahoo.com)
Leila Mossllanejad (mossla_1@yahoo.com)
Alireza gharib (gharibalireza@yahoo.com)
Nahid Dehghan nayeri (dehghann@tums.ac.ir)

Version: 2 Date: 4 May 2014

Author's response to reviews: see over
Cover letter:

The article authors to title “The important factors in clinical competency of Iranian nurses: A Qualitative Study” are committed the following article to print:

1. all authors have agreed to the submission to the journal and that the manuscript is not currently under submission in any other journal
2. All authors approve the content of the manuscript and have contributed significantly to research involved/ the writing of the manuscript;
3. The protocol for the research project has been approved by a suitably constituted research ethics committee of the institution within which the work was undertaken.
4. All participants gave informed consent for the research, and that their anonymity was preserved; and
5. Any financial or personal matters that may pose a conflict of interest.